Author's response to reviews

Title: Induction of protective and therapeutic anti-pancreatic cancer immunity using a reconstructed MUC1 DNA vaccine

Authors:

Yefei Rong (mcyryf@163.com)
Dayong Jin (jin.dayong@zs-hospital.sh.cn)
Wenchuan Wu (wu.wenchuan@zs-hospital.sh.cn)
Wenhui Lou (Lou.wenhui@zs-hospital.sh.cn)
Danshong Wang (Wang.danshong@zs-hospital.sh.cn)
Tiantao Kuang (Kuang.tiantao@zs-hospital.sh.cn)
Xiaoling Ni (Ni.xiaoling@zs-hospital.sh.cn)
Xinyu Qin (Qin.xinyu@zs-hospital.sh.cn)

Version: 5 Date: 11 June 2009

Author's response to reviews:

"Cytotoxicity" in the Y-axis of the fig6 has been changed into "cytotoxicity".